Opinion Dynamics for Utility Maximizing Agents: Exploring the Impact of the Resource Penalty

Prashil Wankhede¹, Nirabhra Mandal², Sonia Martínez² and Pavankumar Tallapragada¹

Abstract—We propose a continuous-time nonlinear model of opinion dynamics with utility-maximizing agents connected via a social influence network. A distinguishing feature of the proposed model is the inclusion of an opinion-dependent resource-penalty term in the utilities, which limits the agents from holding opinions of large magnitude. This model is applicable in scenarios where the opinions pertain to the usage of resources, such as money, time, computational resources etc. Each agent myopically seeks to maximize its utility by revising its opinion in the gradient ascent direction of its utility function, thus leading to the proposed opinion dynamics. We show that, for any arbitrary social influence network, opinions are ultimately bounded. For networks with weak antagonistic relations, we show that there exists a globally exponentially stable equilibrium using contraction theory. We establish conditions for the existence of consensus equilibrium and analyze the relative dominance of the agents at consensus. We also conduct a game-theoretic analysis of the underlying opinion formation game, including on Nash equilibria and on prices of anarchy in terms of satisfaction ratios. Additionally, we also investigate the oscillatory behavior of opinions in a two-agent scenario. Finally, simulations illustrate our findings.

Index Terms—Opinion dynamics, Multi-agent systems, Utility maximization, Game theory.

I. INTRODUCTION

PINION dynamics deals with the modeling and mathematical analysis of how beliefs and ideas evolve and spread within social groups or networks over time. As collective opinions have far-reaching implications in diverse sectors such as policy, public health, sociology, finance or economics, it is paramount to understand their drivers and consequences. Even for smaller groups, comprehending opinion forming is a necessary first step toward the management of mixed multiagent groups, their decision making and subsequent dynamic interactions. While many simpler opinion dynamic models have been developed, modeling and analysis of networked rational agents who react to neighbors' influence remains limited. In this work, we pay particular attention to the setting where the opinions of the agents are related to the usage of resources for a particular task. Each agent has heterogeneous resources available to them which limits their opinions about resource usage. We capture this in our model by including a resource penalty term in the utility functions.

This work was partially supported by Science and Engineering Research Board under grant CRG/2023/008573. ¹Prashil Wankhede and Pavankumar Tallapragada are with the Department of Electrical Engineering, Indian Institute of Science. {{prashilw, pavant}@iisc.ac.in}

²Nirabhra Mandal and Sonia Martínez are with the Department of Mechanical and Aerospace Engineering, University of California, San Diego. {{nmandal, soniamd}@ucsd.edu} *Literature Review:* Many developments in the opinion dynamics literature stem from classical models, including the French-DeGroot (FD) model [1], [2], the Abelson model [3], the Taylor model [4], the Friedkin-Johnsen (FJ) model [5], the Hegselmann-Krause (HK) model [6], the Altafini model [7], and the Deffuant-Weisbuch (DW) model [8].

For an in-depth summary of the above models and survey of the literature, we refer the readers to [9]–[12]. These fundamental models serve as the foundation for many others in the literature. Some examples of recent models include the discrete-time versions of the Altafini model with timevarying signed networks [13], [14]. The work [15] proposes the affine boomerang opinion dynamics model with asynchronous opinion updating, incorporating the phenomenon of the *boomerang* effect into the dynamics. Unlike [7], [13], [14], in this model [15], the opinions do not converge to zero if the network does not satisfy a structural balance property, but rather exhibits bounded fluctuations. Reference [16] proposes the expressed-private-opinion model in which every agent's private opinion is influenced by the expressed opinions of its neighbors. The concept of social power was first introduced in [1] in order to identify the most influential agent in the social network. The DeGroot-Friedkin (DF) model [17] integrates the opinion formation process with social power evolution through a reflected self-appraisal mechanism. A generalized DF model was proposed in [18] and an extension of the DF model to stubborn agents was proposed in [19]. Some works on opinion dynamics [15], [20] also analyze fluctuations and periodic behavior of opinions.

The modeling of opinion dynamics through a gametheoretic or a utility maximization approach is still in its early stages, with only a limited number of works published thus far. Reference [21] showed that cooperative control problems, such as consensus seeking, can be effectively tackled using gametheoretic methods, particularly through potential and weakly acyclic games. The works [22], [23] interpret the FJ-model as best response dynamics within a game-theoretic framework, where each agent aims to minimize a cost function. The work [24] introduced a general framework for social influence grounded in the psychological concept of cognitive dissonance and demonstrated that various opinion dynamics models can be viewed as best response dynamics in a coordination game, where utilities are determined by dissonance functions. In a recent study [25], a dynamic influence maximization game is explored, where a set of competing *players* allocate their fixed resources over certain individuals (who hold opinions about players) to maximize their utilities in the long term. Reference [26] performs a game-theoretical analysis of the asynchronous HK model. The works [27], [28] capture the co-evolution of opinions and actions taken by the agents under a gametheoretic framework. The work [29] utilizes a continuoustime non-linear opinion dynamics model to tune the mutual cooperative behavior of agents in a repeated game. Within this framework, agents make strategic decisions relying upon rationality and reciprocity. Another approach, detailed in [30], introduces a discrete-time opinion dynamics model with a game-theoretic structure where the agents incur a combination of conformity and manipulation costs based on the opinions. The aim of each agent is to minimize this cost.

Exploration of how agents' resources impact their opinions and social influencing capabilities is currently an open question in the opinion dynamics literature. Motivated by this, we adopt a utility maximization and game-theoretic framework in the current work to investigate the limiting effects of agents' resources on their opinions about its usage. A preliminary version of this work appeared in [31], where we assumed the underlying social influence network to be complete. The current work extends [31] to the case of any general social network topologies and also allows for pairwise antagonistic relationships among the agents. We also investigate several new questions, such as exponential stability of the equilibria, price of anarchy and periodic evolution of opinions.

Contributions: The main contributions of this work are:

1) We define agent's utility function to capture the tradeoffs of internal opinion preferences, attachment (or stubbornness) toward its own opinion, conformity or non-conformity and lastly a resource penalty term. The resource penalty term in the utility function depends on the agent's resources, which is a representation of the agent's wealth, time etc. In this work, we assume that agents are involved in the opinion formation process to make some decisions about the use of their resources for a particular task. In the literature, most models of opinion dynamics are generic and seem to assume that the topic of discussion itself has no effect on the opinions' evolution. However, this is not realistic in our setting. Thus, we include a resource penalty term in the utility that keeps the opinions bounded. We propose a novel opinion dynamics model from the utility functions based upon the assumption that every agent myopically seeks to maximize its own utility. We refer to the underlying game as the opinion formation game.

2) Unlike the existing works which consider stubborn agents, under the proposed dynamics agents can reach consensus even if their internal preferences are different. If the agents' opinions reach a consensus equilibrium, we can use the *consensus dominance weights* of the agents to deduce the relative *dominance* of each agent. The consensus dominance weights depend on the resources of the agents.

3) We conduct a game-theoretic analysis of the opinion formation game. Our Nash equilibrium conditions hold even when the network consists of *antagonistic* relationships where opinions of some agents negatively influence the opinions of others. We provide a relation between the set of local Nash equilibria \mathcal{NE}_l , set of Nash equilibria \mathcal{NE} of the opinion formation game and the set of equilibrium points \mathcal{E} of the opinion dynamics. Specifically, we show that $\mathcal{NE} \subseteq \mathcal{NE}_l \subseteq \mathcal{E}$. Further, we give a condition for these sets to coincide.

4) In the case of *weak antagonistic* relationships, we show that the game exhibits a unique Nash equilibrium and the agents' opinions converge to it under the proposed dynamics starting from any arbitrary initial opinion profile. A special case of *weak antagonistic relationships* is one where antagonistic relationships are absent in the social network.

5) In the case where antagonistic relationships are absent, we bound the Price of Anarchy in terms of the *satisfaction ratios* to quantify the inefficiency of the unique Nash equilibrium. The satisfaction ratio of an agent at a given opinion profile of all agents is the ratio of the utility received by the agent at this opinion profile to the maximum possible utility; thus quantifying how "satisfied" that agent is with that opinion profile. Using these bounds, we show that if agents options converge to a consensus then it is a socially optimal outcome.

6) We analyze oscillatory and periodic opinion behavior. We show that it is necessary for the agents to have sufficiently strong antagonistic relations for the two-agent dynamics to have periodic solutions. We also provide sufficient conditions for a Hopf bifurcation to exist for the two-agent dynamics.

Notation: Throughout the paper, we use non-bold letters for denoting scalars, bold lowercase letters for denoting vectors, and bold uppercase letters for denoting matrices. The sets of natural numbers, real numbers, non-negative real numbers and positive real numbers are denoted by $\mathbb{N}, \mathbb{R}, \mathbb{R}_{\geq 0}$ and $\mathbb{R}_{>0}$, respectively. Let $\mathbf{1} \in \mathbb{R}^n$ and $\mathbf{0} \in \mathbb{R}^n$ denote a vector with all elements equal to one and zero, respectively. For any vector $\mathbf{z} \in \mathbb{R}^n$, \mathbf{z}^{\top} denotes its transpose. For any scalar $a \in \mathbb{R}$, |a| denotes its absolute value. For a set S, S^n denotes the Cartesian product of S with itself n times. The empty set is denoted by \emptyset . We denote the difference of any two sets S_1 and S_2 by $S_1 \setminus S_2$. Let $G := (\mathcal{V}, \mathcal{L})$ be a directed graph, where \mathcal{V} is the set of nodes and \mathcal{L} is the set of directed arcs. In a directed graph G, a *directed walk* from a node $i_1 \in \mathcal{V}$ to any node $i_l \in \mathcal{V}$ is a sequence of nodes $i_1 \mapsto i_2 \mapsto \ldots \mapsto i_l$ such that $(i_s, i_{s+1}) \in \mathcal{L}, \forall s \in \{1, 2, \dots, l-1\}.$

Organization of the paper: The rest of the paper is organized as follows. Section II introduces essential preliminaries for subsequent analysis. Section III presents the model for utility functions, and the opinion dynamics, as well as outlines the objectives of the paper. Section IV includes the asymptotic analysis of the model. Section V analyzes consensus equilibria, Nash equilibria and price of anarchy of the opinion formation game. Section VI deals with oscillatory behavior of two-agent opinion dynamics. Section VII includes simulations demonstrating our results. Finally, we conclude in Section VIII. We have included some of the longer proofs in the appendix.

II. PRELIMINARIES

Here, we recall some useful concepts from contraction theory. For a comprehensive description and a compilation of results on contracting dynamical systems, we refer the reader to [32]. First, we define the *induced logarithmic norm* of a matrix and give an interpretation of the same.

Definition 2.1: (Logarithmic norm of a matrix [32]) Given an induced matrix norm $\|.\|$ the induced log-norm of a matrix $\mathbf{A} \in \mathbb{R}^{n \times n}$ is given by

$$\mu(\mathbf{A}) := \lim_{h \to 0^+} \frac{\|\mathbf{I}_n + h\mathbf{A}\| - 1}{h} \in \mathbb{R}.$$

If we use the induced ∞ -matrix norm in the above definition, then we get the induced ∞ -log norm of A as

$$\mu_{\infty}(\mathbf{A}) = \max_{i \in \mathcal{V}} \left(a_{ii} + \sum_{j \in \mathcal{V} \setminus \{i\}} |a_{ij}| \right).$$

The log-norm can be interpreted as the directional derivative of the matrix norm in the direction of **A** evaluated at the identity matrix I_n . It should be noted that the induced log-norm of a matrix is not a matrix norm and can even be negative. This induced log-norm helps us in getting bounds on the norm of the solutions of a continuous time system. For example, consider a continuous-time homogeneous LTI system $\dot{\mathbf{x}} = \mathbf{A}\mathbf{x}$. Using Coppel's inequalities [33], it can be shown that,

$$\|\mathbf{x}(t)\| \le e^{\mu(\mathbf{A})t} \|\mathbf{x}(0)\|$$

.....

If $\mu(\mathbf{A}) < 0$ then the above inequality implies that the solution $\mathbf{x}(t)$ converges to zero exponentially. We can also extend this idea to nonlinear systems. With this motivation, we define a *strongly contracting vector field*.

Definition 2.2: (Strongly contracting vector field [32]) Let $C \subset \mathbb{R}^n$ be convex. Let $\mathbf{f} : C \to \mathbb{R}^n$ be differentiable and let $\mathbf{J}(\mathbf{z})$ denote its Jacobian. Then the vector field \mathbf{f} is said to be *strongly infinitesimally contracting* on C with rate $\alpha > 0$ if

$$\mu(\mathbf{J}(\mathbf{z})) < -\alpha, \,\forall \, \mathbf{z} \in \mathcal{C}.$$

Finally, we recall a result that ensures the existence and uniqueness of an exponentially stable equilibrium for a strongly contracting system. The proof of this theorem is based on the Banach contraction theorem [32].

Theorem 2.3: (Equilibrium of a strongly contracting system [32]) Suppose $C \subset \mathbb{R}^n$ is convex, closed and positive f-invariant. If $\mathbf{f} : C \to \mathbb{R}^n$ is strongly infinitesimally contracting on C with rate $\alpha > 0$ then, \mathbf{f} has a unique globally exponentially stable equilibrium $\mathbf{z}^* \in C$ with global Lyapunov functions $V(\mathbf{z}) = \|\mathbf{z} - \mathbf{z}^*\|$ and $V(\mathbf{z}) = \|\mathbf{f}(\mathbf{z})\|$.

III. MODELING AND PROBLEM SETUP

Consider a set $\mathcal{V} := \{1, \dots, n\}$ of n agents, with heterogeneous resources, that form opinions on a single topic, which is about the quantity of resources to use for a certain purpose. We aim to study the evolution of these opinions as the agents interact with each other over a social network. We start by defining utility functions for each agent. Then, we discuss the utility function and our motivations behind choosing it. Then, we derive the proposed opinion dynamics from the utility functions, assuming that each agent myopically seeks to maximize its utility. Thus, the coupling in the utility functions creates the coupling in the opinion dynamics.

Opinions, Utility Function and its Parameters: We denote the *expressed opinion* of agent $i \in \mathcal{V}$ at time t on the topic as $z_i(t) \in \mathbb{R}$. For brevity, we omit the time argument wherever there is no confusion. The vector $\mathbf{z} := [z_1, \dots, z_n]^\top \in \mathbb{R}^n$ represents the stacked opinions z_i of all agents $i \in \mathcal{V}$. We first present the *utility function* of agent i, then describe the various parameters and provide motivation for the chosen structure.

The complete opinion profile $\mathbf{z} \in \mathbb{R}^n$ determines the utility for each agent $i \in \mathcal{V}$ as follows

$$U_i(\mathbf{z}, p_i) = -\frac{w_i}{2} (z_i - p_i)^2 - \sum_{k \in \mathcal{V}} \frac{a_{ik}}{2} (z_k - z_i)^2 - \frac{1}{4r_i} z_i^4,$$
(1)

where $z_i \in \mathbb{R}$ and $p_i \in \mathbb{R}$ are the expressed opinion and internal *preference* on the topic of agent $i \in \mathcal{V}$ respectively. The parameter $w_i \in \mathbb{R}_{>0}$ is the importance that agent $i \in \mathcal{V}$ attaches to its internal preference on the topic, while $r_i \in \mathbb{R}_{>0}$ is the resources available to agent $i \in \mathcal{V}$ and $a_{ik} \in \mathbb{R}$ is the weight of the influence of agent $k \in \mathcal{V}$ on agent $i \in \mathcal{V}$. We call the three terms in the utility function as the *preference term*, social term and resource penalty, respectively. We make the following standing assumption about the parameters in (1).

(SA1) (*Parameters' signs.*) For each agent $i \in \mathcal{V}$, $p_i \in \mathbb{R}$, $w_i \in \mathbb{R}_{>0}$, $r_i \in \mathbb{R}_{>0}$ and $a_{ik} \in \mathbb{R}$, $\forall k \in \mathcal{V}$.

Note that we allow for heterogeneous agents where each agent can have different parameters in their utility function. Moreover, in this paper, we assume that an agent's utility is affected by others via a directed social influence graph $G := (\mathcal{V}, \mathcal{L}, \mathbf{A})$. The elements of the *adjacency matrix* \mathbf{A} are denoted by $a_{ij} \in \mathbb{R}$. If $a_{ik} \neq 0$ then there exists a directed link from node $k \in \mathcal{V}$ to node $i \in \mathcal{V}$ with link weight a_{ik} . This denotes that agent k's opinion influences the opinion of agent *i*. The sign of a_{ik} denotes the type of influence. Using this idea, we give the following definition.

Definition 3.1: (Neighbor set of an agent.) For each agent $i \in \mathcal{V}$, the set $\mathcal{N}_i^e := \{k \in \mathcal{V} \setminus \{i\} \mid a_{ik} < 0\}$ denotes the set of its *enemies* and the set $\mathcal{N}_i^f := \{k \in \mathcal{V} \setminus \{i\} \mid a_{ik} > 0\}$ denotes the set of its *friends*. Further,

$$\mathcal{N}_i := \mathcal{N}_i^e \cup \mathcal{N}_i^f = \{k \in \mathcal{V} \setminus \{i\} \mid a_{ik} \neq 0\},\$$

denotes the set of neighbors of agent $i \in \mathcal{V}$.

Note that the self loop weights a_{ii} , for any $i \in \mathcal{V}$, do not affect the utility of the agents. We may assume them to be zero without loss of generality. The self influence of the agents is captured by the preference term in the utility function.

Discussion About the Utility Function: Overall, we seek to explain opinion evolution in social networks from a utility maximization perspective. In the sequel, we derive opinion dynamics as a gradient ascent by the agents of their utility functions, with respect to their own opinions. The three terms in (1) represent penalties on agent *i*'s expressed opinion z_i arising from three factors. The preference term penalizes opinion deviation from the internal preference p_i . This penalty is directly proportional to the importance weight (or stubbornness) w_i . The social term penalizes the agent's opinion for being far away from or close to its neighbors' opinions depending upon the type of influence relationships, i.e., on the sign of a_{ik} 's. These two terms in the utility function lead to the first two terms in the opinion dynamics (2), which is the well known Taylor's model of opinion dynamics [4], when only non-antagonistic relations are allowed. In the presence of antagonistic relations, the second term in (1) is motivated by the *boomerang effect* [34]. So, our proposed model differs from the literature primarily in the resource penalty in the utility function (1).

The motivation for the resource penalty comes from the observation that if the topic of discussion is about the resource usage for some purpose, then the resource limitations of the agents must also have an effect on the opinion evolution. In our model, the resource penalty in the utility function restricts agent *i* from holding opinions of larger magnitude. In order to better motivate our model, consider the following example.

Example 3.2: Consider n agents with heterogeneous resource limitations. Let $z_i(t)$, the opinion of agent $i \in \mathcal{V}$ at time t, be the maximum amount of resources agent i is willing to spend on buying a particular product, if it were to do the buying at time t. Agent i has an internal opinion about the value of the product, denoted by p_i . Further, w_i models the confidence of agent i in its internal opinion. In addition, the agents are also influenced by the opinions of others within their social neighborhood. For example, they might look at product reviews or inquire within their social circles, which can influence their opinion about the value of the product. The weights a_{ik} 's model the trust or confidence of agent *i* has in agent k's ability to correctly value the product. Alternatively, these weights could simply model the degree to which agent *i* seeks to mimic or disagree with agent k. However, its spending is ultimately constrained by its resource limitations, which motivates the resource penalty in (1).

The psychological and social effects of resource constraints on decision making of the agents have been studied in various fields such as psychology, economics, transportation etc. We refer the readers to the papers [35], [36] and references therein for more details. For example, [35] discusses the effects of financial constraints on consumer behaviors from four different perspectives: *resource scarcity, choice restriction, social comparison* and *environmental uncertainity*.

Remark 3.3: (On the Resource Penalty.) The parameter r_i denotes the amount of the resources (such as wealth, time etc.) available to agent $i \in \mathcal{V}$. It could be a proxy for the maximum budget an agent has to spend, such as the amount of money available to buy a certain good, the amount of time or other resources available for doing a task etc. In our work, we assume that the amount of resource r_i is static in time. This is reasonable to assume in scenarios where the agents' opinions are about the usage of their resources itself. The agents actually use their resources only after its opinion converges or after a sufficiently long evolution time of its opinion. Thus, we can think of the resource penalty as a soft constraint on their opinions, which are in turn about how much resources they could expend for a good or a task. The greater the resources that agent *i* has, the larger the magnitude of the opinions it can hold.

In this work, we choose a quartic resource penalty function; however, a more general class of penalty functions is also acceptable. Mathematically, all that is needed for ensuring the boundedness of opinions is to choose a non-negative resource penalty term that dominates the other terms in the utility (1) for large enough $|z_i|$.

The following is a standing assumption in this paper.

(SA2) (*Preferences are fixed parameters.*) For each agent $i \in \mathcal{V}, p_i \in \mathbb{R}$ is a fixed parameter.

The results can easily be extended for the case where $p_i = z_i(0)$, for each agent $i \in \mathcal{V}$. For example, we can let $\mathbf{p} = \mathbf{z}(0)$ and consider (\mathbf{z}, \mathbf{p}) as the state variables with $\dot{\mathbf{p}} = 0$. Since this provides little additional value, we choose to think of \mathbf{p} as a fixed parameter. In the case where $\mathbf{p} = \mathbf{z}(0)$, we can study the dependence of the opinion evolution on the initial opinions $\mathbf{z}(0)$ by carrying out a parametric study.

Opinion Dynamics: We assume that at each time instant, agent $i \in \mathcal{V}$ revises its opinion by doing a gradient ascent of its utility function U_i , given in (1), with respect to its own opinion z_i . Thus, for each $i \in \mathcal{V}$, we have

$$\dot{z}_i = -w_i[z_i - p_i] + \sum_{k \in \mathcal{V}} a_{ik}[z_k - z_i] - \frac{z_i^3}{r_i}.$$
 (2)

We can rewrite (2) equivalently as

$$\dot{z}_i = f_i(\mathbf{z}, p_i) := S_i(z_i, p_i) + C_i(\mathbf{z}), \quad \forall i \in \mathcal{V},$$
(3a)

$$S_i(z_i, p_i) := -w_i [z_i - p_i] - \frac{z_i^3}{r_i},$$
 (3b)

$$C_i(\mathbf{z}) := \sum_{k \in \mathcal{V}} a_{ik} [z_k - z_i].$$
(3c)

Note that $\forall i \in \mathcal{V}$, the *self function* $S_i(\cdot, p_i)$ depends only on i's own opinion, its preference p_i and other parameters. On the other hand, $\forall i \in \mathcal{V}$, the *crowd function* $C_i(\cdot)$ depends on the deviations of i's opinion z_i from its neighbors' opinions z_k 's. If $a_{ik} > 0$ (< 0) then $i \in \mathcal{V}$ wants to agree (disagree) with agent k and hence the term in $C_i(\cdot)$ corresponding to agent k drives z_i towards (away from) the opinion of $k \in \mathcal{V}$.

Now, we make some observations about the self function $S_i(\cdot, p_i)$, which hold $\forall i \in \mathcal{V}$. $S_i(\cdot, p_i)$ is continuous and strictly decreasing with $\lim_{z_i \to -\infty} S_i(z_i, p_i) = \infty$ and $\lim_{z_i \to +\infty} S_i(z_i, p_i) = -\infty$. Thus, $S_i(\cdot, p_i)$ has exactly one real root for every fixed value of $p_i \in \mathbb{R}$. Let us denote the real root of $S_i(\cdot, p_i)$ as $m_i(p_i) \in \mathbb{R}$, *i.e.*, $S_i(m_i(p_i), p_i) = 0$. Hereafter, we will exclude the preference argument in $S_i(\cdot, p_i)$ and $m_i(\cdot)$ wherever there is no confusion. Moreover, by considering $S_i(0)$ and $S_i(p_i)$, we can verify that $0 \leq |m_i| \leq |p_i|$ and $m_i p_i \geq 0$. $m_i \in \mathbb{R}$ can be interpreted as the opinion that agent i would attain under (2) if it were *socially closed* from the influence of other agents, *i.e.* $a_{ik} = 0$, $\forall k \in \mathcal{V}$.

Opinion Dynamics in Absence of Antagonistic Relations: An important special case is one where there are no antagonistic relations, which we formally state in the following assumption.

(A3) (*No antagonistic relations.*) $\forall i \in \mathcal{V}, \mathcal{N}_i^e = \emptyset$. • Under Assumption (A3), opinion dynamics (3a) reduces to

$$\dot{z}_i = f_i(\mathbf{z}, p_i) := S_i(z_i) + C_i^+(\mathbf{z}), \quad \forall i \in \mathcal{V},$$
(4)

where
$$C_i^+(\mathbf{z}) = \begin{cases} 0 & ; \text{ if } \mathcal{N}_i = \emptyset, \\ \sum_{i \in \mathcal{N}} a_{ij} \left[\bar{z}_i - z_i \right] & ; \text{ otherwise} \end{cases}$$
 (5)

١

$$\bar{z}_i := \sum_{k \in \mathcal{V}} \frac{a_{ik}}{\left(\sum_{j \in \mathcal{V}} a_{ij}\right)} z_k \,. \tag{6}$$

Note that, under Assumption (A3), for any $i \in \mathcal{V}$, if $\mathcal{N}_i \neq \emptyset$ then \overline{z}_i in (6) is well defined. Moreover, notice that this \overline{z}_i is a convex combination of the opinions z_k of agent *i*'s neighbors. So in (4), it suffices $\forall i \in \mathcal{V}$ to know only \overline{z}_i and be unaware of other individual agents' opinions. Finally, note that $\forall i \in \mathcal{V}$,

$$S_{i}(z_{i}) \begin{cases} > 0, & z_{i} < m_{i} \\ = 0, & z_{i} = m_{i}, \ C_{i}^{+}(\mathbf{z}) \\ < 0, & z_{i} > m_{i} \end{cases} \begin{cases} > 0, & z_{i} < \bar{z}_{i} \\ = 0, & z_{i} = \bar{z}_{i} \\ < 0, & z_{i} > \bar{z}_{i}. \end{cases}$$
(7)

Objectives: The important analytical questions regarding the opinion dynamics (2) that we study in this paper include

- asymptotic properties
- existence of a consensus equilibrium and its properties
- characterization of Nash equilibria and their relation to the equilibria of the dynamics (2)
- · bounds on price of anarchy

• oscillatory behavior of opinions in the two agent case. We also study these questions for the cases outlined in Assumption (A4) and (A3) to provide stronger results.

IV. ASYMPTOTIC BEHAVIOR OF OPINIONS

In this section, we study the asymptotic behavior of opinions. We show ultimate boundedness of opinions and provide conditions that guarantee existence and uniqueness of a globally exponentially stable equilibrium point. We denote the set of equilibrium points of (2) as a function of preferences **p** as

$$\mathcal{E}(\mathbf{p}) := \{ \mathbf{z} \in \mathbb{R}^n \, | \, \dot{\mathbf{z}} = \mathbf{f}(\mathbf{z}, \mathbf{p}) = \mathbf{0} \}.$$
(8)

Hereafter, we drop the preference argument in $\mathcal{E}(\mathbf{p})$ wherever there is no confusion. We first guarantee that, under the dynamics (2), irrespective of the initial opinion profile, the opinions never grow unbounded and are in fact, ultimately bounded. The ultimate boundedness of opinions is a consequence of the resource penalty term in (1), which heavily penalizes any agent for holding opinions of greater magnitude.

Theorem 4.1: (Ultimate boundedness of opinions.) Let $\mathbf{z}(t)$ be the solution to (2) from the initial condition $\mathbf{z}(0)$. Then $\exists \eta \ge 0$ (independent of $\mathbf{z}(0)$) and $\exists T(\mathbf{z}(0)) \ge 0$ such that $|z_i(t)| \le \eta, \forall i \in \mathcal{V}, \forall t \ge T$. Additionally, $\exists \Omega \subseteq \mathbb{R}^n$ which is convex, compact and positive invariant under (2).

The existence of an ultimate bound can be shown using similar arguments involved in the proof of [31, Theorem 3.1]. Using the same arguments, we can also show existence of a convex, compact sublevel set Ω of the quadratic Lyapunov function $V(\mathbf{z}) := 0.5 \mathbf{z}^{\top} \mathbf{z}$ such that $\dot{V}(\mathbf{z}) < 0$ outside Ω and the solution with any initial condition $\mathbf{z}(0) \in \mathbb{R}^n$ enters the set Ω in finite time $\bar{T}(\mathbf{z}(0))$. We skip the proof for brevity.

Note that one can explicitly find an ultimate bound η but we skip it for brevity. Also note that Theorem 4.1 holds for all values of the parameters and hence does not depend on the type of interactions among agents or the structure of the social network. From the proof of Theorem 4.1, we note that for any initial condition, the solution is uniformly bounded over all time. Then the local Lipschitzness of the vector field in (2) can be used to show existence and uniqueness of solutions of (2)for all time. Next, we characterize the conditions under which the opinions converge to an equilibrium. Recall that for each $i \in \mathcal{V}, S_i(\cdot)$ has a unique root at m_i , *i.e.* $S_i(m_i) = 0$. Using this, we show that the opinion of a socially closed agent (if one exists) converges to its corresponding m_i . The next result can be shown using (7) and the fact that the dynamics (2) for every such agent reduces to the scalar differential equation $\dot{z}_i = S(z_i)$. We skip the proof for brevity.

Lemma 4.2: (Convergence of opinions of socially closed agents.) Consider the dynamics given by (3a). Let $\mathbf{z}(t)$ denote the solution of (2) from an initial condition $\mathbf{z}(0)$. Then, $\lim_{t\to\infty} z_i(t) = m_i$; $\forall i \in \mathcal{V}_{cl}$, where \mathcal{V}_{cl} is the set of socially closed agents, i.e., $\mathcal{V}_{cl} := \{i \in \mathcal{V} \mid \mathcal{N}_i = \emptyset\} \subseteq \mathcal{V}$. We next address the general case, where there may be some socially closed agents and some agents influenced by others. In this case, the parameters, such as importance weights, resources and inter-agent influence weights, all play a role in determining whether there is an equilibrium point, its uniqueness and its stability. Using Theorem 2.3, we present a sufficient condition for (2) to have a unique globally exponentially stable equilibrium. We begin by stating an assumption.

(A4) (Weak antagonistic relations.) For each agent $i \in \mathcal{V}$, $w_i > \sum_{k \in \mathcal{N}_i^e} 2|a_{ik}|$, where \mathcal{N}_i^e is the set of enemies of i. • Notice that under Assumption (SA1), $w_i > 0, \forall i \in \mathcal{V}$. Thus, the case of no antagonistic relations stated in Assumption (A3) is a special case of Assumption (A4). All the results which hold under Assumption (A4) also hold under Assumption (A3). Now, we present the main result of this section. Its proof is in the appendix.

Theorem 4.3: (Existence and uniqueness of equilibrium points.) Consider the dynamics given by (2). Suppose Assumption (A4) holds. Then, there exists a unique globally exponentially stable equilibrium point $z^* \in \mathcal{E}$.

Remark 4.4: (On the weak antagonistic relationships condition.) The condition of Assumption (A4) means that the willingness of every agent $i \in \mathcal{V}$ to hold an opinion close to its internal preference is greater than twice the aggregate influence weight of its enemies. Under this assumption, Theorem 4.3 shows that the opinions converge to a unique equilibrium.

Based on Lemma 4.2 and Theorem 4.3, we can give a stronger result for ultimate boundedness and on the location of the unique equilibrium in case there are no antagonistic relations among agents. We deal with this case next.

Convergence of opinions in the absence of antagonistic relations: In the absence of antagonistic relationships among agents, it is possible to give an ultimate bound that is more intuitive and easier to compute using the unique roots of the self functions $(m_i \text{ such that } S_i(m_i) = 0, i \in \mathcal{V})$. We can also say that the unique equilibrium in this case is in a specific set defined by m_i 's, for $i \in \mathcal{V}$. Let,

$$m_{\min}(\mathbf{p}) := \min\{m_i\}_{i \in \mathcal{V}}, \ m_{\max}(\mathbf{p}) := \max\{m_i\}_{i \in \mathcal{V}}, \ (9)$$

and the corresponding interval

from $j \in \mathcal{V} \setminus \mathcal{V}_{\min}$ to *i*.

$$\mathcal{M}(\mathbf{p}) := [m_{\min}(\mathbf{p}), m_{\max}(\mathbf{p})].$$
(10)

For brevity, we will exclude the preference argument in (9) and (10) wherever there is no confusion. We are now ready to show that the opinions converge to \mathcal{M}^n (proof in Appendix).

Proposition 4.5: (Convergence to the set \mathcal{M}^n in absence of antagonistic relations.) Consider the opinion dynamics given by (2). Suppose Assumption (A3) holds. Let m_{\min} , m_{\max} and \mathcal{M} be as defined in (9) and (10) respectively. Then, \mathcal{M}^n is positively invariant under the opinion dynamics (2). Let $\mathbf{z}(t)$ be the solution to (2) from an initial condition $\mathbf{z}(0) \in \mathbb{R}^n$. Then $\mathbf{z}(t)$ converges to \mathcal{M}^n . Further, define $\mathcal{V}_{\max} := \{i \in \mathcal{V} \mid m_i = m_{\min}\}$ Suppose $m_{\min} < m_{\max}$. Then the following statements are equivalent.

(i) The unique equilibrium z* ∈ E lies in the interior of Mⁿ and ∃T(z(0)) ≥ 0 such that z(t) ∈ Mⁿ, ∀t ≥ T(z(0)).
(ii) ∀i ∈ V_{max}, ∃ a directed walk in G starting from j ∈ V \ V_{max} to i and ∀i ∈ V_{min}, ∃ a directed walk in G starting

Under Assumption (A3), Proposition 4.5 guarantees that the set \mathcal{M}^n is positively invariant. Moreover, the unique globally exponentially stable equilibrium point z^* that Theorem 4.3 guarantees lies in \mathcal{M}^n . Thus, we can immediately guarantee that $\mathbf{z}(t)$ converges to \mathcal{M}^n , possibly asymptotically. If $m_{\min} <$ $m_{\rm max}$ then only agents belonging to sets $\mathcal{V}_{\rm max}$ and $\mathcal{V}_{\rm min}$ can have equilibrium opinions at the boundary of the set \mathcal{M} . For the unique equilibrium to be in the interior of \mathcal{M}^n , it is both necessary and sufficient that every agent in the sets \mathcal{V}_{max} and \mathcal{V}_{\min} is directly or indirectly influenced by at least one agent in sets $\mathcal{V} \backslash \mathcal{V}_{\max}$ and $\mathcal{V} \backslash \mathcal{V}_{\min},$ respectively. Note that this condition is satisfied if the social network G is strongly connected. Finally, if the unique equilibrium z^* lies in the interior of \mathcal{M}^n then solutions converge to \mathcal{M}^n in finite time. In this case, the ultimate bound \mathcal{M}^n has an additional advantage that it depends only on the m_i 's (whose interpretation is provided in Section III) and hence can be computed easily using the parameters w_i, p_i and r_i . Finally, note that if $m_{\text{max}} = m_{\text{min}}$ then the opinions of agents in absence of antagonistic relations always converge to a unique consensus equilibrium.

V. CONSENSUS AND NASH EQUILIBRIA

In this section, we analyze consensus equilibria of the opinion dynamics (2) and Nash equilibria of the underlying game. We also explore the relation between the Nash equilibrium set of the underlying game and the set of equilibria of the opinion dynamics (2). Finally, we also analyze the price of anarchy.

A. Consensus Equilibria

First, we deal with the consensus equilibria of the model, i.e., equilibria of the form $\xi \mathbf{1}$, with $\xi \in \mathbb{R}$. We refer to the case of $\xi = 0$ as a *neutral consensus* since all the agents have neutral opinions (equal to 0) in this case. On the other hand, we refer to the case of $\xi \neq 0$ as a non-neutral consensus. In the following lemma, we present conditions for (2) to have a consensus equilibrium. We use the form of the dynamics in (3a) and the functions in (3) to justify our claims.

Theorem 5.1: (Necessary and sufficient conditions for existence of a consensus equilibrium.) Consider the dynamics (2) (equivalently (3a)). For each $i \in \mathcal{V}$, let $m_i \in \mathbb{R}$ be the unique point such that $S_i(m_i) = 0$. Then, $\xi \mathbf{1} \in \mathcal{E}$ if and only if $m_i = \xi, \forall i \in \mathcal{V}$.

Proof: If $z_i = \xi$, $\forall i \in \mathcal{V}$, for some $\xi \in \mathbb{R}$ then $C_i(\xi \mathbf{1}) = 0$, $\forall i \in \mathcal{V}$. Hence, from (3a), $\xi \mathbf{1} \in \mathcal{E}$ iff $S_i(\xi) = 0, \forall i \in \mathcal{V}$. Since m_i is the unique root of $S_i(.)$, the claim follows.

Remark 5.2: (Consensus formation among agents.) Theorem 5.1 states that it is both necessary and sufficient for all the m_i 's to be the same for the opinion dynamics model to have a consensus equilibrium. It is evident that if the agents are to arrive at a consensus equilibrium, then all their preferences p_i 's must be of the same sign. When $p_i = 0$, $\forall i \in \mathcal{V}$, then the only possible consensus equilibrium is the neutral consensus, *i.e.*, every agent reaches a neutral opinion on the topic. If the preferences of the agents have different signs, then the opinions of agents can never reach an exact consensus in equilibrium. However, other equilibria that are arbitrarily close to consensus may still exist.

Now, if the weak antagonistic relationships condition given in Assumption (A4) holds and there exists a consensus equilibrium, then, from Theorem 4.3, it is the only equilibrium of the dynamics and the agents always achieve consensus starting from any initial opinion vector.

When the agents attain a consensus equilibrium, we can measure how much influence an agent has on the whole group by measuring the deviation of the consensus value from its preference. Note that if $p_i = 0$, for some $i \in \mathcal{V}$, then $m_i = 0$ and hence the only consensus equilibrium possible (if it exists) is neutral. So we consider $p_i \neq 0$, $\forall i \in \mathcal{V}$ to give the next result on dominance and discuss it in the remark following it.

Proposition 5.3: (Consensus deviation from preference.) Consider the dynamics (2) or equivalently (3a). Suppose that $p_i \neq 0, \forall i \in \mathcal{V}$. For each agent $i \in \mathcal{V}$, let us define $\sigma_i := w_i r_i$ and $\Delta_i(\xi) := |p_i - \xi|$. If $\xi \mathbf{1} \in \mathcal{E}$, with $\xi \in \mathbb{R}$, then $\sigma_i \Delta_i(\xi) = \sigma_j \Delta_j(\xi), \forall i, j \in \mathcal{V}$. In particular, $\sigma_i > \sigma_j$ iff $\Delta_i(\xi) < \Delta_j(\xi)$. *Proof:* Since $p_i \neq 0, \forall i \in \mathcal{V}$, we also have $0 < |m_i| < |p_i| \forall i \in \mathcal{V}$. Then by Theorem 5.1, $\xi \neq 0$. Further, from Theorem 5.1, we get $r_i S_i(\xi) = 0, \forall i \in \mathcal{V}$, which then implies

$$\sigma_i(p_i - \xi) = \xi^3 = \sigma_j(p_j - \xi), \ \forall i, j \in \mathcal{V}.$$

Since $\sigma_i > 0$, $\forall i \in \mathcal{V}$, the result now follows.

Remark 5.4: (Dominance in consensus.) Let us call $\forall i \in \mathcal{V}$, the scalar σ_i as the consensus dominance weight of the agent *i*. Suppose all agents have a non-neutral preference and they attain consensus at $\xi \in \mathbb{R}$. Then, Proposition 5.3 states that if an agent $i \in \mathcal{V}$ has higher consensus dominance weight than agent $j \in \mathcal{V}$, then ξ is closer to *i*'s preference than that of *j*. Note that the consensus dominance weight is directly proportional to the importance weight an agent assigns to its preference and the resources available to it. Thus, an agent with very high resources can exert more influence even if it gives less importance weight to its internal preference. On the other hand, if an agent has lower resources, then it has to have much higher importance weight to have more influence in the group.

Next, we study the relation between equilibria of the opinion dynamics (2) and Nash equilibria of the underlying game.

B. Nash Equilibria

Here we carry out a Nash Equilibrium analysis of the opinion formation game. Recall that every agent is interested in maximizing its utility U_i given in (1) by suitably choosing its opinion z_i . Thus, this gives a *strategic form game* $\mathcal{G} = \langle \mathcal{V}, (\mathbb{R})_{i \in \mathcal{V}}, (U_i)_{i \in \mathcal{V}} \rangle$ among the set of agents \mathcal{V} , with the *strategy* of agent $i \in \mathcal{V}$ being its opinion $z_i \in \mathbb{R}$ and its utility function being $U_i(\cdot)$. For the sake of convenience, we let \mathbf{z}_{-i} denote the opinions of all agents other than *i*. Then, the set of *Nash equilibria* of the game \mathcal{G} is

$$\mathcal{NE}(\mathbf{p}) := \{ \mathbf{z}^* \in \mathbb{R}^n \mid \forall i \in \mathcal{V}, \\ U_i(z_i^*, \mathbf{z}_{-i}^*, p_i) \ge U_i(z_i, \mathbf{z}_{-i}^*, p_i), \forall z_i \in \mathbb{R} \}.$$
(11)

Note that for a Nash equilibrium \mathbf{z}^* , z_i^* is agent *i*'s best response over all opinions $z_i \in \mathbb{R}$ to \mathbf{z}_{-i}^* , the opinion profile of all the other agents. However, in the dynamics (2), each agent updates its opinion according to the gradient ascent of its utility with respect to its opinion while assuming that the

other agents do not change their opinions. Hence, the agents at each time instant revise their opinion only towards the "local" best response. This motivates the next definition.

Definition 5.5: (Local Nash equilibrium.) A strategy profile $\mathbf{z}^* \in \mathbb{R}^n$ is said to be a local Nash equilibrium if and only if $\forall i \in \mathcal{V}, \exists \rho_i \in \mathbb{R}_{>0}$ such that

$$U_i(z_i^*, \mathbf{z}_{-i}^*, p_i) \ge U_i(z_i, \mathbf{z}_{-i}^*, p_i), \ \forall z_i \text{ s.t. } |z_i^* - z_i| \le \rho_i$$

The set of local Nash equilibria of \mathcal{G} is denoted by $\mathcal{NE}_l(\mathbf{p})$.

For simplicity, we will exclude the preference arguments in $\mathcal{NE}(\mathbf{p})$ and $\mathcal{NE}_l(\mathbf{p})$ wherever there is no confusion. It is easy to see that a Nash equilibrium is also a local Nash equilibrium and hence $\mathcal{NE} \subseteq \mathcal{NE}_l$. In the next result, we show that every local Nash equilibrium is an equilibrium point of (2).

Lemma 5.6: (Local Nash equilibrium is an equilibrium of the opinion dynamics.) If an opinion profile z^* is such that $z^* \in \mathcal{NE}_l$ then $z^* \in \mathcal{E}$.

Proof: Since $\mathbf{z}^* \in \mathcal{NE}_l$, it implies that $\forall i \in \mathcal{V}, z_i^*$ locally maximizes $U_i(\cdot, \mathbf{z}_{-i}^*)$. Thus, the partial derivative of $U_i(\cdot)$ with respect to z_i evaluated at $(z_i^*, \mathbf{z}_{-i}^*)$ is zero. The claim then follows immediately from (2) and (8).

Lemma 5.6 states that every local Nash equilibrium of \mathcal{G} is also an equilibrium point of dynamics (2). But the converse need not be true. In the following result, we give conditions for an opinion profile $\mathbf{z}^* \in \mathbb{R}^n$ that is an equilibrium point of the opinion dynamics to be a local Nash equilibrium of the opinion formation game.

Theorem 5.7: (Conditions for an equilibrium point of (2) to be a local Nash equilibrium.) Consider the dynamics (2) and the set of equilibrium points \mathcal{E} in (8). Let $\mathbf{z}^* = (z_i^*, \mathbf{z}_{-i}^*) \in \mathcal{E}$. Then, $\mathbf{z}^* \in \mathcal{NE}_l$ only if for each $i \in \mathcal{V}$,

$$(z_i^*)^2 \ge \tau_i := \frac{r_i}{3} \left[\sum_{k \in \mathcal{N}_i^e} |a_{ik}| - \sum_{k \in \mathcal{N}_i^f} a_{ik} - w_i \right].$$
(12)

Moreover, if inequality (12) is strict $\forall i \in \mathcal{V}$, then $\mathbf{z}^* \in \mathcal{NE}_l$. *Proof:* Let the hypothesis be true. From Definition 5.5 we know that $\mathbf{z}^* = (z_i^*, \mathbf{z}_{-i}^*)$ is a local Nash equilibrium if and only if $\forall i \in \mathcal{V}, z_i^*$ locally maximizes $U_i(\cdot, \mathbf{z}_{-i}^*)$. Now since $\mathbf{z}^* \in \mathcal{E}$, by the definitions in (2) and (8), it is clear that for each $i \in \mathcal{V}, z_i^*$ satisfies the first-order necessary conditions for it to be a local maximizer of $U_i(\cdot, \mathbf{z}_{-i}^*)$.

Now, suppose that $\mathbf{z}^* \in \mathcal{NE}_l$. Then, we have that

$$\left. \frac{\partial^2}{\partial z_i^2} U_i(z_i, \mathbf{z}_{-i}^*) \right|_{z_i^*} = \frac{3}{r_i} \left[\tau_i - (z_i^*)^2 \right] \le 0, \quad \forall i \in \mathcal{V} \,. \tag{13}$$

This proves the necessary condition in (12). Finally, note that if (13) is a strict inequality for a $\mathbf{z}^* \in \mathcal{E}$, then by the second order sufficiency condition for a point to be a local maximizer, z_i^* is a local maximizer of $U_i(\cdot, \mathbf{z}_{-i}^*)$ for each $i \in \mathcal{V}$.

The statement of the previous result can be combined with the result in Theorem 4.1 to provide a condition for which the opinion formation game does not have any local Nash equilibrium. We state this in the next result, proof of which is intuitive since no equilibrium point of (2) can exist beyond any ultimate bound (which always exists).

Corollary 5.8: (Non-existence of local Nash equilibria.) Suppose $\eta_i > 0$ is an ultimate bound on z_i for any $i \in \mathcal{V}$ under (2). If there exists $i \in \mathcal{V}$ such that $\tau_i > \eta_i^2$, with τ_i defined in (12), then $\mathcal{NE}_l = \emptyset$.

Finally, to end this subsection, we provide a sufficient condition under which the different equilibria sets (\mathcal{E} , \mathcal{NE} and \mathcal{NE}_l) are equal. Additionally, we also give a sufficient condition under which the opinion formation game has a unique Nash equilibrium. We state this in the following result.

Theorem 5.9: (Equality of equilibria sets and uniqueness of Nash equilibrium.) Consider the dynamics in (3a) and the set of equilibrium points \mathcal{E} in (8). Suppose for each agent $i \in \mathcal{V}, \tau_i \leq 0$. Then, $\mathcal{NE} = \mathcal{NE}_l = \mathcal{E}$. Moreover, if Assumption (A4) holds then it is a singleton set.

Proof: Suppose that $\tau_i \leq 0, \forall i \in \mathcal{V}$. First, it is obvious that $\mathcal{NE} \subseteq \mathcal{NE}_l$ and Lemma 5.6 implies that $\mathcal{NE}_l \subseteq \mathcal{E}$. Next, we show that $\mathcal{E} \subseteq \mathcal{NE}_l$. Consider an equilibrium $\mathbf{z}^* \in \mathcal{E}$. It can be easily verified that for each $i \in \mathcal{V}, U_i(\cdot, \mathbf{z}^*_{-i})$ is strictly concave in z_i . Since $\mathbf{z}^* \in \mathcal{E}$, the first order necessary condition for a point to be a local maximizer of $U_i(\cdot, \mathbf{z}^*_{-i})$ is satisfied for each i. Hence, $\mathbf{z}^* \in \mathcal{NE}_l$. Finally, we show that $\mathcal{NE}_l \subseteq \mathcal{NE}$. Since $\forall i \in \mathcal{V}, U_i(\cdot, \mathbf{z}^*_{-i})$ is a strictly concave function for each \mathbf{z}^*_{-i} , the implication follows directly from the definitions of \mathcal{NE} and \mathcal{NE}_l . This completes the proof of the first part of the claim. Now suppose that Assumption (A4) holds. Then again $\tau_i < 0, \forall i \in \mathcal{V}$. Uniqueness of Nash equilibrium now follows from Theorem 4.3 and first part of the claim.

Remark 5.10: We can interpret w_i as the weight of influence of a stubborn virtual agent on agent *i*. Using this interpretation, let us define the weighted in-degree for each agent $j \in \mathcal{V}$ in social network G as $d_j^{in} := \left| w_j + \sum_{k \in \mathcal{V} \setminus \{j\}} a_{jk} \right|.$ Then for each $j \in \mathcal{V}$, $\tau_j = -r_j d_j^{in}/3$ and $\tau_j < 0 (= 0)$ if and only if $d_j^{in} > 0 (= 0)$. The condition $d_j^{in} > 0$, $\forall i \in \mathcal{V}$ can be interpreted as follows: For each agent $i \in \mathcal{V}$, the aggregate influence weight of its friends (including the virtual fully stubborn agent since $w_i > 0$ is greater than the aggregate influence weight of its enemies. If this holds true then Theorem 5.9 states that each equilibrium point of the dynamics is also a Nash equilibrium of the game. Next, if Assumption (A4) holds then for each $j \in \mathcal{V}, d_j^{in} > 0$ and hence $\tau_i < 0$. In other words, the opinion formation game has a unique Nash equilibrium if the willingness of each agent to be close to its internal preference (or stubbornness) is more than twice the aggregate influence weight of its enemies. Moreover, thanks to Theorem 4.3, we can say that the opinions under (2) always converge to this unique Nash equilibrium, starting from any initial condition.

C. Price of Anarchy

Next, we analyze the *price of anarchy* of the game underlying the opinion dynamics (2). In the entirety of this subsection, we will use cost minimization perspective rather than utility maximization. We let the cost incurred by agent $i \in \mathcal{V}$ for opinion profile \mathbf{z} be $\chi_i(\mathbf{z}, \mathbf{p}) := -U_i(\mathbf{z}, \mathbf{p})$, where $U_i(\mathbf{z}, \mathbf{p})$ is given by (1). In order to ensure non-negativity of the prices of anarchy defined below, we need $\chi_i(\mathbf{z}, \mathbf{p}) \ge 0$; $\forall \mathbf{z} \in \mathbb{R}^n$. Hence, in this subsection we will assume that all inter-agent relations are non-antagonistic, i.e., Assumption (A3) holds. We consider price of anarchy for two most commonly used objective functions in the game theory literature, namely, the egalitarian and the utilitarian costs

$$\mathsf{C}_E(\mathbf{z},\mathbf{p}) := \max_{i \in \mathcal{V}} \chi_i(\mathbf{z},\mathbf{p}), \quad \mathsf{C}_U(\mathbf{z},\mathbf{p}) := \sum_{i \in \mathcal{V}} \chi_i(\mathbf{z},\mathbf{p}),$$

to measure the inefficiency of Nash equilibria of the game underlying the opinion dynamics (2). Hereafter, we will again exclude the preference arguments in $\chi_i(\cdot, \cdot)$, $C_E(\cdot, \cdot)$ and $C_U(\cdot, \cdot)$ for brevity. In the following definitions of price of anarchy, we will assume $\mathbf{p} \neq \mathbf{0}$ to ensure the positivity of $C_E(\mathbf{z})$ and $C_U(\mathbf{z})$ at any $\mathbf{z} \in \mathbb{R}^n$. This ensures that prices of anarchy (14) are well defined. We discuss the case when $\mathbf{p} = \mathbf{0}$ in a later remark.

Definition 5.11: (**Price of anarchy**.) Consider the opinion formation game \mathcal{G} corresponding to the opinion dynamics (2), and \mathcal{NE} , the set of its Nash equilibria. Suppose Assumption (A3) holds and $\mathbf{p} \neq \mathbf{0}$. The egalitarian and the utilitarian prices of anarchy, π_e and π_u , respectively are defined as

$$\pi_e := \frac{\max_{\mathbf{z} \in \mathcal{N}\mathcal{E}} \mathsf{C}_E(\mathbf{z})}{\min_{\mathbf{z} \in \mathbb{R}^n} \mathsf{C}_E(\mathbf{z})} \ge 1; \quad \pi_u := \frac{\max_{\mathbf{z} \in \mathcal{N}\mathcal{E}} \mathsf{C}_U(\mathbf{z})}{\min_{\mathbf{z} \in \mathbb{R}^n} \mathsf{C}_U(\mathbf{z})} \ge 1.$$
(14)

In Definition 5.11, π_e compares the cost incurred by the worst performing agent at the worst Nash equilibrium to the minimum possible cost C_E . Similarly, π_u compares the total cost incurred by all agents at the worst Nash equilibrium to the minimum possible cost C_U . The closer the value of the prices of anarchy (14) is to unity, the better the quality of the Nash equilibrium. We now define the *satisfaction ratio* at opinion profile z for every agent $i \in \mathcal{V}$ in order to compare the cost incurred by agent i at z and the minimum possible cost it could incur. Using these ratios, we can give some upper bounds on the prices of anarchy (14). In order to keep the satisfaction ratios well defined, we need positivity of $\chi_i(z)$ everywhere. Hence we assume the following,

(A5) (Non-zero preferences.) $\forall i \in \mathcal{V}, p_i \neq 0.$

Definition 5.12: (Satisfaction ratio.) Consider the cost function $\chi_i(\mathbf{z})$ of each agent $i \in \mathcal{V}$. Suppose Assumptions (A3) and (A5) hold. The satisfaction ratio SR_i for any agent $i \in \mathcal{V}$ at any $\mathbf{z} \in \mathbb{R}^n$ is defined as,

$$\mathsf{SR}_{i}(\mathbf{z}) := \frac{\chi_{i}(\mathbf{z})}{\min_{\mathbf{z} \in \mathbb{R}^{n}} \chi_{i}(\mathbf{z})} \ge 1.$$
(15)

The next result shows that if there are no antagonistic relations among the agents, then for each agent $i \in \mathcal{V}$, $\chi_i(\cdot)$ is convex. Note that this result says $\chi_i(\mathbf{z})$ is convex in \mathbf{z} and not just with respect to z_i . Proof of the following result is in the appendix.

Lemma 5.13: (Convexity of cost function.) Suppose that Assumption (A3) holds. Then for each $i \in \mathcal{V}, \chi_i(\cdot)$ is convex with $m_i \mathbf{1}$ as one of its minimizer.

Now, from Theorem 4.3 and Theorem 5.9, we know that in the absence of antagonistic relations, the game has a unique Nash equilibrium $z^* \in \mathcal{NE}$. In the next result, whose proof is in the Appendix, we use the satisfaction ratios to give an upper bound on π_e and π_u .

Theorem 5.14: (**Bounds on prices of anarchy**.) Consider the dynamics (2). Suppose Assumptions (A3) and (A5) hold. Let us denote the unique Nash equilibrium of the game underlying the opinion dynamics (2) by $\mathbf{z}^* \in \mathcal{NE}$. Then,

$$1 \le \pi_e \le \max_{i \in \mathcal{V}} \mathsf{SR}_i(\mathbf{z}^*) \ , \ 1 \le \pi_u \le \sum_{i \in \mathcal{V}} \mathsf{SR}_i(\mathbf{z}^*) \ , \quad (16)$$

with π_e and π_u as defined in Definition 5.11.

A consequence of this result is that, for the special case where the unique Nash equilibrium is a consensus equilibrium, $\pi_e = \pi_u = 1$, i.e., consensus equilibrium (if it exists) is socially optimal. We formally state and prove this next.

Corollary 5.15: (PoA is unity for a non-neutral consensus equilibrium.) Consider the dynamics (2). Suppose Assumptions (A3) and (A5) hold. Suppose the unique $\mathbf{z}^* \in \mathcal{NE} = \mathcal{E}$ is a non-neutral consensus equilibrium, i.e., $\mathbf{z}^* = m\mathbf{1}$ for some $m \neq 0$. Then, $\pi_e = \pi_u = 1$.

Proof: From Theorem 5.1, Lemma 5.13 and Definition 5.12, it follows that, $SR_i(m1) = 1$, $\forall i \in \mathcal{V}$. From (16), we thus have $\pi_e = 1$. From the proof of Theorem 5.14 for the bound on π_u , we have

$$1 \le \pi_u \le \frac{\sum_{i \in \mathcal{V}} \chi_i(m\mathbf{1})}{\sum_{i \in \mathcal{V}} \chi_i(m\mathbf{1})} = 1,$$

since $\mathbf{z}^* = m\mathbf{1}$ and $m_i = m, \forall i \in \mathcal{V}$.

Note that neutral and non-neutral consensus equilibrium exists only if $\mathbf{p} = \mathbf{0}$ and Assumption (A5) holds, respectively. Consensus equilibrium cannot exist if $\exists i, j \in \mathcal{V}$ with $p_i = 0$ and $p_j \neq 0$. Corollary 5.15 states that non-neutral consensus is socially optimal. In the next remark, we argue that neutral consensus is also socially optimal.

Remark 5.16: (Neutral consensus is socially optimal.) Let Assumption (A3) hold and $\mathbf{p} = \mathbf{0}$. Theorems 5.1, 4.3 and 5.9 imply that the unique equilibrium point $\mathbf{z}^* = \mathbf{0} \in \mathcal{E} = \mathcal{N}\mathcal{E}$. Now, from Lemma 5.13 we know that $\mathbf{0}$ is a minimizer of $\chi_i(\mathbf{z})$, $C_E(\mathbf{z})$ and $C_U(\mathbf{z})$. Thus, prices of anarchy (14) and satisfaction ratios (15) are not defined in this case. But since $C_E(\mathbf{z}^*) = \min_{\mathbf{z} \in \mathbb{R}^n} C_E(\mathbf{z}) = 0 = C_U(\mathbf{z}^*) = \min_{\mathbf{z} \in \mathbb{R}^n} C_U(\mathbf{z})$, neutral consensus equilibrium is also socially optimal.

VI. OSCILLATORY BEHAVIOR OF OPINIONS

In [31], we studied a special case of the dynamics proposed in the current paper. We showed through numerical simulations that in certain scenarios, the opinions exhibit oscillatory behavior. In this section, we analyze such behavior in the case of a pair of agents $\mathcal{V} = \{1, 2\}$. For a general social network with n agents, the analysis is significantly more complicated due to the higher dimension of the state space but also due to the complexity added by the social network. Thus, the analysis in the general case of a social network with n agents would have to be a separate research work in itself and is out of the scope of this paper. Even though our analysis is restricted to two agents, it is still relevant since it can help us understand opinion behaviors or decision making in important systems such as two-party politics, duopoly economic markets etc.

The opinion dynamics (2) for two agents is,

$$\dot{z}_1 = S_1(z_1) + a_{12}[z_2 - z_1], \ \dot{z}_2 = S_2(z_2) - a_{21}[z_2 - z_1].$$
 (17)

For the *two-agent dynamics* (17), it can be easily verified that if the opinions $z_1(t)$ and $z_2(t)$ exhibit oscillatory behavior then they will have the same fundamental period of oscillations. We state this in the following result (proof in Appendix).

Lemma 6.1: (*Equal period of oscillations*) Consider the two-agent dynamics (17). Suppose the opinions $z_1(t)$ and $z_2(t)$ exhibit oscillations with fundamental periods $T_1 > 0$ and $T_2 > 0$, respectively. Then, $T_1 = T_2$.

Note that periodic orbits can exist for (17) only if at least one of the agents has an antagonist influence on the other as otherwise Theorem 4.3 guarantees convergence of opinions to the unique equilibrium from any arbitrary initial condition. In the next result (proof in Appendix), we give a necessary condition for the existence of periodic orbits for (17).

Lemma 6.2: (Necessary condition for existence of periodic solutions) Consider the two-agent dynamics (17). Opinions $z_1(t)$ and $z_2(t)$ exhibit periodic behavior only if all the following hold: (i) $a_{12} < 0$ or $a_{21} < 0$, (ii) $a_{12}a_{21} \neq 0$, and (iii) $(a_{21} + w_1 + w_2 + a_{12}) < 0$.

Remark 6.3: (Geometric and other interpretations of the conditions for the existence of periodic opinions.) The condition given in Lemma 6.2 ensures existence of the following ellipse in the phase plane,

$$\frac{z_1^2}{r_1} + \frac{z_2^2}{r_2} = -\frac{(w_1 + w_2 + a_{12} + a_{21})}{3} =: v > 0.$$
(18)

From the proof of Lemma 6.2, we see that the $div(\mathbf{f}(\mathbf{z}))$ is equal to zero on the ellipse and positive (negative) inside (outside) the ellipse. Thus, any periodic orbit of (17) in the phase plane should necessarily intersect the above ellipse.

The oscillatory behavior of opinions appears only when at least one of the agents has an antagonist influence on the other. This is similar to the so-called *boomerang effect* [34], where at least one of the agents has an antagonist influence on another agent. As a result, the agent on whom there is an antagonist influence shifts its opinion away from the other agent. In such situations, the opinions of the agents could converge to a disagreement equilibrium or could possibly keep on oscillating forever never converging to an equilibrium.

Now that we have dealt with necessary conditions for a periodic solution of (17) to exist, we conclude this section by giving sufficient conditions for the agents to exhibit periodic opinion profiles. More specifically, we give sufficient conditions on the conformity weights a_{12} , a_{21} for a Hopf bifurcation to exist (proof in Appendix).

Theorem 6.4: (Existence of Hopf bifurcation) For the two-agent dynamics (17) with $a_{21} \in \mathbb{R}$ as the bifurcation parameter, let $\kappa_i := \left[w_i + \frac{3m_i^2}{r_i}\right] > 0, i \in \{1, 2\}$. Let $m_1 = m_2 = m \in \mathbb{R}$ and $\left[a_{12}(\kappa_2 - \kappa_1) - \kappa_1^2\right] > 0$. Then, a family of periodic orbits of (17) bifurcates from the consensus equilibrium $\mathbf{z}^* = m\mathbf{1}$ at $a_{21} = a_{21}^* := -(\kappa_1 + \kappa_2 + a_{12})$.

VII. SIMULATIONS

In this section, we present some simulations to demonstrate our analytical results. All the simulation results were generated using MATLAB and the ODE 45 solver. Through out this section, the *i*th element of any parameter data vector corresponds to agent *i* and we use \approx (=) signs for indicating approximate (exact) values of the provided data.

In the first set of simulations, we consider a group of 6 agents forming opinions according to (2). We assume that the agents are connected via an influence network that is shown in Figure 1. The matrix of influence weights A is equal to the adjacency matrix associated with the graph shown in Figure 1. In Figure 2a, we illustrate the case where the opinions of all 6 agents reach a non-neutral consensus equilibrium, with consensus value equal to 40. The model parameters used to



Fig. 1: A social network consisting of 6 agents. The direction of any link denotes the direction of influence and the number near arrowhead of any directed link (k, i) represents the corresponding link weight a_{ik} .

simulate this case are as follows. The vectors of initial opinions $z_i(0)$, importance weights w_i , resources r_i , agents' internal preferences p_i of all six agents are

$$\begin{split} \mathbf{z}_0 &\approx \begin{bmatrix} 43.90 & 36.34 & 49.00 & 30.69 & 38.77 & 37.63 \end{bmatrix}, \\ \mathbf{w} &\approx \begin{bmatrix} 1.09 & 2.98 & 2.59 & 1.82 & 1.01 & 2.65 \end{bmatrix}, \\ \mathbf{r} &\approx \begin{bmatrix} 317.43 & 814.54 & 789.07 & 852.26 & 505.64 & 635.66 \end{bmatrix}, \\ \mathbf{p} &\approx \begin{bmatrix} 225.22 & 66.40 & 71.33 & 81.16 & 165.39 & 77.99 \end{bmatrix}, \end{split}$$

respectively. For these parameters, $m_i = 40$, $\forall i \in \mathcal{V}$, which is equal to the consensus value. Thus, this simulation verifies Theorem 5.1. For each agent $i \in \mathcal{V}$, let z_i^{∞} denote its asymptotic opinion value. The vector whose each element is the absolute difference between an agent's final consensus opinion and its preference opinion $(|z_i^{\infty} - p_i|)$ and the consensus dominance weights σ_i are

$$\mathbf{d} \approx \begin{bmatrix} 185.22 & 26.40 & 31.33 & 41.16 & 125.39 & 37.99 \end{bmatrix} \\ \boldsymbol{\sigma} \approx \begin{bmatrix} 345.5 & 2424.4 & 2042.5 & 1555.1 & 510.4 & 1684.8 \end{bmatrix}.$$

From this data, we can verify that the dominance claim in Proposition 5.3 is satisfied in this case.



Fig. 2: Convergence of opinions. (a) Consensus equilibrium. (b) Disagreement equilibrium within \mathcal{M}^n .

Figure 2b depicts the scenario where the opinions of the 6 agents with no antagonistic relations converge to a disagreement equilibrium in the compact set \mathcal{M}^n . The model parameters in this case are,

$$\mathbf{z}_0 \approx \begin{bmatrix} -69.02 & -28.03 & -46.22 & 24.01 & -23.92 & 38.38 \end{bmatrix}$$

$$\mathbf{w} \approx \begin{bmatrix} 0.22 & 3.48 & 0.56 & 2.56 & 0.95 & 2.62 \end{bmatrix}$$

$$\mathbf{r} \approx \begin{bmatrix} 317.43 & 814.54 & 789.07 & 852.26 & 505.64 & 635.66 \end{bmatrix}$$

$$\mathbf{p} \approx \begin{bmatrix} -18.31 & 18.92 & -12.43 & 6.68 & 3.46 & 7.00 \end{bmatrix}.$$

The agents do not achieve consensus because

$$\mathbf{m} \approx \begin{vmatrix} -8.78 & 17.14 & -10.10 & 6.56 & 3.38 & 6.81 \end{vmatrix}$$

which violates the necessary condition for consensus given in Theorem 5.1. From Figure 2b, it can be seen that every agent's opinion converges to a value in the set $\mathcal{M} \approx [-10.10, 17.14]$ which verifies the results of Theorem 4.3 and Proposition 4.5.

Figure 3a demonstrates the oscillatory behavior exhibited by opinions of two agents under (17). The model parameters for this case are, $\mathbf{z_0} \approx |-0.0349 - 0.0039|, [a_{12} a_{21}] \approx$ $\begin{bmatrix} -6.6667 & 3.3267 \end{bmatrix}$, $\mathbf{w} \approx \begin{bmatrix} 2 & 1.3333 \end{bmatrix}$, $\mathbf{p} = \begin{bmatrix} 0 & 0 \end{bmatrix}$ and $\mathbf{r} = \begin{bmatrix} 10 & 5 \end{bmatrix}$. Consider (17) with the above parameter values and let a_{21} be the bifurcation parameter. Notice that since $p_1 = p_2 = 0$, neutral consensus $\mathbf{z}^* = \mathbf{0} \in \mathcal{E}$, $\forall a_{21} \in \mathbb{R}$. These values satisfy the assumptions of Theorem 6.4 ensuring the existence a Hopf bifurcation at $a_{21}^* = 3.3333$. The eigenvalues of the Jacobian $J(\mathbf{z}^*, a_{21}^*)$ are $\pm 0.6667j$. As a result, a family of periodic orbits bifurcates out of the neutral consensus equilibrium at a_{21}^* and the periodic solution shown in Figure 3a is a member of this family corresponding to $a_{21} = 3.3267 < a_{21}^*$. Figure 3b shows the corresponding trajectory in the phase plane intersecting the ellipse defined in (18). These simulations support the claim of Lemma 6.2and its interpretation made in Remark 6.3.



Fig. 3: Oscillatory behavior of opinions. (a) Opinion trajectory. (b) Intersection of corresponding trajectory with the ellipse.

VIII. CONCLUSIONS

We proposed a non-linear model of opinion dynamics to capture the effect of heterogeneous resources available to the agents on their opinions. In contrast to our prior work [31], we dealt with general social networks with (possibly) antagonistic relations. We showed ultimate boundedness of opinions and provided sufficient conditions for the dynamics to have a globally exponentially stable equilibrium point. We also provided necessary and sufficient condition for the existence of a consensus equilibrium and quantified social dominance at consensus. Further, we showed that the set of Nash equilibria of the opinion formation game is a subset of the set of equilibrium points of the dynamics and provided conditions for these two sets to coincide. In the absence of antagonistic relations, we gave stronger results. We quantified the quality of the Nash equilibria with respect to two commonly used prices of anarchy (PoA), provided bounds on these PoA's in terms of the satisfaction ratios and proved that converging to a consensus equilibrium is a socially optimal outcome. Finally, we analyzed the periodic behavior of opinions exhibited

by the proposed dynamics for the case of two agents. We provided necessary conditions for periodic solutions to exist and sufficient conditions for a Hopf bifurcation to occur at the consensus equilibrium. Future research directions include extensions of the model to a multi-topic scenario, analysis of periodic behavior exhibited by opinions in the presence of antagonistic relations for a general *n*-agent case, and exploration of a more general class of utility functions and resource penalty functions.

APPENDIX

A. Proof of Theorem 4.3

To prove this, we show that the dynamics (2) is strongly contracting. Let $\mathbf{J}(\mathbf{z}) := \begin{bmatrix} \frac{\partial \mathbf{f}(\mathbf{z})}{\partial \mathbf{z}} \end{bmatrix}$ denote the Jacobian matrix of (2). The $(i, j)^{\text{th}}$ element of $\mathbf{J}(\mathbf{z})$ is

$$[\mathbf{J}(\mathbf{z})]_{ij} = \begin{cases} -w_i - \left(\sum_{k \in \mathcal{V} \setminus \{i\}} a_{ik}\right) - \frac{3z_i^2}{r_i}, & \text{if } i = j; \\ a_{ij}, & \text{if } i \neq j. \end{cases}$$

The induced ∞ -log norm of $\mathbf{J}(\mathbf{z})$ is:

$$\mu_{\infty}(\mathbf{J}(\mathbf{z})) = \max_{i \in \mathcal{V}} \left[-w_i + \sum_{k \in \mathcal{N}_i^e} 2|a_{ik}| - \frac{3z_i^2}{r_i} \right]$$

Under Assumption (A4), it can be seen that $\exists k > 0$ such that $\mu_{\infty}(\mathbf{J}(\mathbf{z})) < -k$; $\forall \mathbf{z} \in \mathbb{R}^n$. Now, Theorem 4.1 guarantees existence of a set $\Omega \subset \mathbb{R}^n$ which is convex, closed and positively invariant under the dynamics (2). Further, from the discussion below Theorem 4.1, we know that no equilibrium exists outside Ω . Thus, from Theorem 2.3, the dynamics (2) is strongly contracting in Ω and the unique equilibrium is globally exponentially stable. This completes the proof.

B. Proof of Proposition 4.5

Notice that under Assumption (A3), the dynamics (2) reduces to (4). First, we prove positive invariance of \mathcal{M}^n under (4) by inspecting the vector field at \mathbf{z} on the boundary of \mathcal{M}^n . Consider any $i \in \mathcal{V}$ such that $z_i = m_{\max}$. From (7), we see that $S_i(z_i) \leq 0$ and $C_i^+(\mathbf{z}) \leq 0$, as $m_i \leq m_{\max}$ and \bar{z}_i is a convex combination of $\{z_j\}_{j=1}^n$ and $\mathbf{z} \in \mathcal{M}^n$. Thus, $f_i(\mathbf{z}) \leq 0$. Similarly, for any $i \in \mathcal{V}$ such that $z_i = m_{\min}$, $f_i(\mathbf{z}) \geq 0$. This implies that \mathcal{M}^n is positively invariant under (4).

Now, we show that $\mathbf{z}(t)$ converges to \mathcal{M}^n . Note that Assumption (A3) is a special case of Assumption (A4) and hence Theorem 4.3 guarantees that there is a unique globally exponentially stable equilibrium point $\mathbf{z}^* \in \mathcal{E}$. We know that under Assumption (A3), \bar{z}_i is a convex combination of $\{z_k\}_{k\in\mathcal{N}_i}$ and that (7) holds. So, $\forall \mathbf{z} \notin \mathcal{M}^n$, $\exists i \in \mathcal{V}$ such that $\dot{z}_i \neq 0$. Thus, the unique globally exponentially stable equilibrium point $\mathbf{z}^* \in \mathcal{M}^n$ and hence $\mathbf{z}(t)$ converges to \mathcal{M}^n .

Next, we prove the last claim in the result, i.e., for the special case of $m_{\min} < m_{\max}$. We have already seen that under Assumption (A3), the unique globally exponentially stable equilibrium point $\mathbf{z}^* \in \mathcal{M}^n$. The following statement is a direct consequence of (7).

(S1) For $i \in \mathcal{V}$, $z_i^* = m_{\max}$ if and only if $m_i = m_{\max}$ and $z_j^* = m_{\max}$, $\forall j \in \mathcal{N}_i$.

From the necessary condition on m_i in (S1), we can say that (S2) $z_i^* < m_{\max}, \forall i \in (\mathcal{V} \setminus \mathcal{V}_{\max}).$

A further consequence of (S1) is that for $i \in \mathcal{V}_{\max}$, if $\exists j \in \mathcal{N}_i \cap (\mathcal{V} \setminus \mathcal{V}_{\max})$ then $\bar{z}_i^* < m_{\max}$ and hence $z_i^* < m_{\max}$. This fact along with (S1) and (S2) imply that for $i \in \mathcal{V}_{\max}$, $z_i^* < m_{\max}$ if and only if there is a directed walk in G from $j \in (\mathcal{V} \setminus \mathcal{V}_{\max})$ to *i*. We can make similar observations corresponding to m_{\min} . From all these observations, we can finally say that

(S3) \mathbf{z}^* lies in the interior of \mathcal{M}^n if and only if $\forall i \in \mathcal{V}_{\max}$, \exists a directed walk in G starting from $j \in \mathcal{V} \setminus \mathcal{V}_{\max}$ to i and $\forall i \in \mathcal{V}_{\min}$, \exists a directed walk in G starting from $j \in \mathcal{V} \setminus \mathcal{V}_{\min}$ to i.

Convergence to \mathcal{M}^n occurs in finite time if and only if the globally exponentially stable equilibrium \mathbf{z}^* lies in the interior of \mathcal{M}^n . This completes the proof of the result.

C. Proof of Lemma 5.13

Suppose Assumption (A3) holds. Then $\forall i \in \mathcal{V}$, it can be verified using Gerschgorin disc theorem that the Hessian of $\chi_i(\mathbf{z})$ is positive semidefinite, $\forall \mathbf{z} \in \mathbb{R}^n$. Hence, $\chi_i(\mathbf{z})$ is convex. Now, $m_i \mathbf{1}$ satisfies the first order necessary condition for a point to be a local minimizer of $\chi_i(\mathbf{z})$. Since $\chi_i(\mathbf{z})$ is convex, $m_i \mathbf{1}$ is also a minimizer of $\chi_i(\mathbf{z})$.

D. Proof of Theorem 5.14

We get the lower bound on π_e and π_u from the Definition 5.11. To get an upper bound on π_e , note that

$$\min_{\mathbf{z}\in\mathbb{R}^n}\max_{i\in\mathcal{V}}\chi_i(\mathbf{z})\geq \max_{i\in\mathcal{V}}\min_{\mathbf{z}\in\mathbb{R}^n}\chi_i(\mathbf{z}).$$

Using this inequality and Lemma 5.13, we can upper bound π_e defined in Definition 5.11 as, $\forall j \in \mathcal{V}$,

$$\pi_e \leq \frac{\max_{i \in \mathcal{V}} \chi_i(\mathbf{z}^*)}{\max_{i \in \mathcal{V}} \chi_i(m_i \mathbf{1})} \leq \max_{i \in \mathcal{V}} \left[\frac{\chi_i(\mathbf{z}^*)}{\chi_j(m_j \mathbf{1})} \right],$$

for any $j \in \mathcal{V}$. Choosing j = i, Definition 5.12 and Lemma 5.13 implies the bound on π_e given in (16). Now, consider π_u as in Definition 5.11. From Lemma 5.13, we have,

$$\min_{\mathbf{z}\in\mathbb{R}^n}\sum_{i\in\mathcal{V}}\chi_i(\mathbf{z})\geq\sum_{i\in\mathcal{V}}\chi_i(m_i\mathbf{1}).$$

Using this we get, $\forall j \in \mathcal{V}$,

$$\pi_u \leq \frac{\sum_{i \in \mathcal{V}} \chi_i(\mathbf{z}^*)}{\sum_{i \in \mathcal{V}} \chi_i(m_i \mathbf{1})} = \sum_{i \in \mathcal{V}} \left[\frac{\chi_i(\mathbf{z}^*)}{\sum_{j \in \mathcal{V}} \chi_j(m_j \mathbf{1})} \right] \leq \sum_{i \in \mathcal{V}} \frac{\chi_i(\mathbf{z}^*)}{\chi_j(m_j \mathbf{1})}$$

Choosing j = i, Definition 5.12 and Lemma 5.13 implies the bound on π_u given in (16).

E. Proof of Lemma 6.1

To prove this, rewrite (17) as,

$$\dot{z}_1 - S_1(z_1) + a_{12}z_1 = a_{12}z_2, \ \dot{z}_2 - S_2(z_2) + a_{21}z_2 = a_{21}z_1$$

Now, suppose $z_1(t)$ and $z_2(t)$ are periodic with fundamental periods $T_1 > 0$ and $T_2 > 0$ respectively. Then, $S_1(z_1(t))$ and \dot{z}_1 (resp. $S_2(z_2(t))$ and \dot{z}_2) are also periodic with fundamental period T_1 (resp. T_2). Then, we can see that $\exists m, n \in \mathbb{N}$ such that $T_1 = m T_2$ and $T_2 = n T_1$. Thus, m = n = 1.

F. Proof of Lemma 6.2

From the discussion just above Lemma 6.2, it is necessary that at least one of a_{12} or $a_{21} < 0$ for periodic solutions to exist. Further, it is also necessary that both influence weights are non-zero. Otherwise, from Lemma 4.2, the socially closed agent's opinion would converge to its own m_i value. The opinion of other agent then would always converge to an equilibrium. Now, it remains to show that $(a_{21}+w_1+w_2+a_{12}) < 0$ is necessary for existence of periodic solutions. We prove the inverse using Bendixson's criteria. Note the divergence of the vector field (17) at $\mathbf{z} \in \mathbb{R}^2$ is,

$$\operatorname{div}(\mathbf{f}(\mathbf{z})) = -(w_1 + w_2 + a_{12} + a_{21}) - \frac{3z_1^2}{r_1} - \frac{3z_2^2}{r_2}.$$

Suppose $(a_{21} + w_1 + w_2 + a_{12}) > 0$. Then, div $(\mathbf{f}(\mathbf{z})) < 0$, $\forall \mathbf{z} \in \mathbb{R}^2$. Thus, by Bendixson's criteria [37, Lemma 2.2] there are no periodic orbits lying entirely in \mathbb{R}^2 .

Now, consider the case when $(a_{21} + w_1 + w_2 + a_{12}) = 0$. In this case, div $(\mathbf{f}(\mathbf{z})) < 0$, $\forall \mathbf{z} \in \mathbb{R}^2 \setminus \{\mathbf{0}\}$ and div $(\mathbf{f}(\mathbf{z})) = 0$ iff $\mathbf{z} = \mathbf{0}$. Suppose there exists a periodic orbit, then in the simply connected region $S \subset \mathbb{R}^2$ enclosed by the periodic orbit, $\iint_{S} (\operatorname{div}(\mathbf{f}(\mathbf{z}))) dz_1 dz_2 = 0$. However, this cannot happen unless $S = \{\mathbf{0}\}$, which cannot correspond to a periodic solution. Hence, again, periodic orbits cannot exist.

G. Proof of Theorem 6.4

Under the assumption $m_1 = m_2 = m$, it follows from Theorem 5.1 that $\mathbf{z}^* = m\mathbf{1} \in \mathcal{E}$; $\forall a_{21} \in \mathbb{R}$. The Jacobian of (17) evaluated about \mathbf{z}^* is,

$$\mathbf{J}(\mathbf{z}^*, a_{21}) = \begin{bmatrix} (-\kappa_1 - a_{12}) & a_{12} \\ a_{21} & (-\kappa_2 - a_{21}) \end{bmatrix}$$

Under the stated assumptions, if $a_{21} = a_{21}^*$ then the eigenvalues of $\mathbf{J}(\mathbf{z}^*, a_{21}^*)$ are purely imaginary. This satisfies the first assumption of the Hopf bifurcation theorem [38, Theorem 3.4.2]. The eigenvalues $\lambda(a_{21})$ of $\mathbf{J}(\mathbf{z}^*, a_{21})$ which are purely imaginary at $a_{21} = a_{21}^*$ vary smoothly with the a_{21} . For values of a_{21} sufficiently close to a_{21}^* , the real part of complex conjugate eigenvalue pair can be given as $Re(\lambda(a_{21})) = -0.5 [\kappa_1 + \kappa_2 + a_{12} + a_{21}]$. Then, the derivative of $Re(\lambda(a_{21}))$ with respect to a_{21} evaluated at a_{21}^* is $\frac{d}{da_{21}} [Re(\lambda(a_{21}))]|_{a_{21}=a_{21}^*} = -0.5 \neq 0$. Hence, the second assumption of the Hopf theorem is also satisfied. The claim now follows directly from the Hopf bifurcation theorem.

REFERENCES

- [1] J. R. French Jr, "A formal theory of social power." *Psychological Review*, vol. 63, no. 3, p. 181, 1956.
- [2] M. H. DeGroot, "Reaching a consensus," Journal of the American Statistical Association, vol. 69, no. 345, pp. 118–121, 1974.
- [3] R. P. Abelson, "Mathematical models of the distribution of attitudes under controversy," *Contributions to Mathematical Psychology*, 1964.
- [4] M. Taylor, "Towards a mathematical theory of influence and attitude change," *Human Relations*, vol. 21, no. 2, pp. 121–139, 1968.
- [5] N. E. Friedkin and E. C. Johnsen, "Social influence and opinions," *Journal of Mathematical Sociology*, vol. 15, no. 3-4, pp. 193–206, 1990.
- [6] R. Hegselmann and U. Krause, "Opinion dynamics and bounded confidence models, analysis, and simulation," *Journal of Artificial Societies* and Social Simulation, vol. 5, no. 3, 2002.
- [7] C. Altafini, "Consensus problems on networks with antagonistic interactions," *IEEE Transactions on Automatic Control*, vol. 58, no. 4, pp. 935–946, 2013.

- [8] G. Deffuant, D. Neau, F. Amblard, and G. Weisbuch, "Mixing beliefs among interacting agents," *Advances in Complex Systems*, vol. 3, no. 01n04, pp. 87–98, 2000.
- [9] A. V. Proskurnikov and R. Tempo, "A tutorial on modeling and analysis of dynamic social networks. Part I," *Annual Reviews in Control*, vol. 43, pp. 65–79, 2017.
- [10] —, "A tutorial on modeling and analysis of dynamic social networks. Part II," Annual Reviews in Control, vol. 45, pp. 166–190, 2018.
- [11] B. D. Anderson, F. Dabbene, A. V. Proskurnikov, C. Ravazzi, and M. Ye, "Dynamical networks of social influence: Modern trends and perspectives," *IFAC-PapersOnLine*, vol. 53, no. 2, pp. 17616–17627, 2020.
- [12] Q. Zha, G. Kou, H. Zhang, H. Liang, X. Chen, C.-C. Li, and Y. Dong, "Opinion dynamics in finance and business: a literature review and research opportunities," *Financial Innovation*, vol. 6, pp. 1–22, 2020.
- [13] Z. Meng, G. Shi, K. H. Johansson, M. Cao, and Y. Hong, "Behaviors of networks with antagonistic interactions and switching topologies," *Automatica*, vol. 73, pp. 110–116, 2016.
- [14] J. Liu, X. Chen, T. Başar, and M. A. Belabbas, "Exponential convergence of the discrete-and continuous-time altafini models," *IEEE Transactions* on Automatic Control, vol. 62, no. 12, pp. 6168–6182, 2017.
- [15] P. Cisneros-Velarde, K. S. Chan, and F. Bullo, "Polarization and fluctuations in signed social networks," *IEEE Transactions on Automatic Control*, vol. 66, no. 8, pp. 3789–3793, 2021.
- [16] M. Ye, Y. Qin, A. Govaert, B. D. Anderson, and M. Cao, "An influence network model to study discrepancies in expressed and private opinions," *Automatica*, vol. 107, pp. 371–381, 2019.
- [17] P. Jia, A. Mirtabatabaei, N. E. Friedkin, and F. Bullo, "Opinion dynamics and the evolution of social power in influence networks," *SIAM Review*, vol. 57, no. 3, pp. 367–397, 2015.
- [18] M. Ye and B. D. Anderson, "Modelling of individual behaviour in the degroot-friedkin self-appraisal dynamics on social networks," in 2019 18th European Control Conference (ECC), 2019, pp. 2011–2017.
- [19] Y. Tian, P. Jia, A. MirTabatabaei, L. Wang, N. E. Friedkin, and F. Bullo, "Social power evolution in influence networks with stubborn individuals," *IEEE Transactions on Automatic Control*, vol. 67, no. 2, pp. 574–588, 2022.
- [20] A. Bizyaeva, A. Franci, and N. E. Leonard, "Sustained oscillations in multi-topic belief dynamics over signed networks," in 2023 American Control Conference (ACC), 2023, pp. 4296–4301.
- [21] J. R. Marden, G. Arslan, and J. S. Shamma, "Cooperative control and potential games," *IEEE Transactions on Systems, Man, and Cybernetics, Part B (Cybernetics)*, vol. 39, no. 6, pp. 1393–1407, 2009.
- [22] J. Ghaderi and R. Srikant, "Opinion dynamics in social networks with stubborn agents: Equilibrium and convergence rate," *Automatica*, vol. 50, no. 12, pp. 3209–3215, 2014.
- [23] D. Bindel, J. Kleinberg, and S. Oren, "How bad is forming your own opinion?" Games and Economic Behavior, vol. 92, pp. 248–265, 2015.
- [24] P. Groeber, J. Lorenz, and F. Schweitzer, "Dissonance minimization as a microfoundation of social influence in models of opinion formation," *The Journal of Mathematical Sociology*, vol. 38, no. 3, pp. 147–174, 2014.
- [25] S. R. Etesami, "Open-loop equilibrium strategies for dynamic influence maximization game over social networks," *IEEE Control Systems Letters*, vol. 6, pp. 1496–1500, 2022.
- [26] S. R. Etesami and T. Başar, "Game-theoretic analysis of the Hegselmann-Krause model for opinion dynamics in finite dimensions," *IEEE Transactions on Automatic Control*, vol. 60, no. 7, pp. 1886–1897, 2015.
- [27] L. Zino, M. Ye, and M. Cao, "A two-layer model for coevolving opinion dynamics and collective decision-making in complex social systems," *Chaos: An Interdisciplinary Journal of Nonlinear Science*, vol. 30, no. 8, p. 083107, 2020.
- [28] H. D. Aghbolagh, M. Ye, L. Zino, Z. Chen, and M. Cao, "Coevolutionary dynamics of actions and opinions in social networks," *IEEE Transactions* on Automatic Control, vol. 68, no. 12, pp. 7708–7723, 2023.
- [29] S. Park, A. Bizyaeva, M. Kawakatsu, A. Franci, and N. E. Leonard, "Tuning cooperative behavior in games with nonlinear opinion dynamics," *IEEE Control Systems Letters*, vol. 6, pp. 2030–2035, 2022.
- [30] S. R. Etesami, S. Bolouki, A. Nedić, T. Başar, and H. V. Poor, "Influence of conformist and manipulative behaviors on public opinion," *IEEE Transactions on Control of Network Systems*, vol. 6, no. 1, pp. 202– 214, 2019.
- [31] P. Wankhede, N. Mandal, and P. Tallapragada, "Opinion dynamics for utility maximizing agents with heterogeneous resources," in 2023 American Control Conference (ACC), 2023, pp. 1054–1061.
- [32] F. Bullo, Contraction Theory for Dynamical Systems, 1.1 ed. Kindle Direct Publishing, 2023. [Online]. Available: https://fbullo.github.io/ctds

- [33] W. A. Coppel, *Stability and asymptotic behavior of differential equations*, ser. Heath mathematical monographs. Heath, 1965.
- [34] R. P. Abelson and J. C. Miller, "Negative persuasion via personal insult," *Journal of Experimental Social Psychology*, vol. 3, no. 4, pp. 321–333, 1967.
- [35] R. W. Hamilton, C. Mittal, A. Shah, D. V. Thompson, and V. Griskevicius, "How financial constraints influence consumer behavior: An integrative framework," *Journal of Consumer Psychology*, vol. 29, no. 2, pp. 285–305, 2019.
- [36] C. Ho and C. Mulley, "Intra-household interactions in tour-based mode choice: The role of social, temporal, spatial and resource constraints," *Transport Policy*, vol. 38, pp. 52–63, 2015.
- [37] H. K. Khalil, Nonlinear systems; 3rd ed. Upper Saddle River, NJ: Prentice-Hall, 2002.
- [38] J. Guckenheimer and P. Holmes, Nonlinear oscillations, dynamical systems, and bifurcations of vector fields. Springer, 2013.



Prashil Wankhede (Student Member IEEE) received the B.Tech. degree in Electrical Engineering from Sardar Patel College of Engineering, Mumbai, India, in 2021. He is currently a Ph.D. student in the Electrical Engineering Department at the Indian Institute of Science, Bangalore, India. His research interests include multi-agent systems, opinion dynamics, game theory and non-linear control.



Nirabhra Mandal (Student Member IEEE) received the B.Tech. degree in Electrical Engineering from the Institute of Engineering and Management, Kolkata, India, in 2017 and the M.Tech.(Res) degree in Electrical Engineering from the Indian Institute of Science, Bengaluru, India, in 2021. He is currently a Ph.D. student in the Mechanical and Aerospace Engineering Department at University of California, San Diego, USA. His research interests include multiagent systems, distributional robust optimzation, popu-

lation games, evolutionary dynamics on networks, and nonlinear control.



Sonia Martínez (Fellow IEEE) is a Professor of Mechanical and Aerospace Engineering at the University of California, San Diego, CA, USA. She received her Ph.D. degree in Engineering Mathematics from the Universidad Carlos III de Madrid, Spain, in May 2002. She was a Visiting Assistant Professor of Applied Mathematics at the Technical University of Catalonia, Spain (2002- 2003), a Postdoctoral Fulbright Fellow at the Coordinated Science Laboratory of the University of Illinois, Urbana-Champaign (2003-

2004) and the Center for Control, Dynamical systems and Computation of the University of California, Santa Barbara (2004-2005). Her research interests include the control of networked systems, multi-agent systems, nonlinear control theory, and planning algorithms in robotics. She became a Fellow of IEEE in 2018. She is a co-author (together with F. Bullo and J. Cortés) of "Distributed Control of Robotic Networks" (Princeton University Press, 2009). She is a co-author (together with M. Zhu) of "Distributed Optimization-based Control of Multi-agent Networks in Complex Environments" (Springer, 2015). She is the Editor in Chief of the recently launched CSS IEEE Open Journal of Control Systems.



Pavankumar Tallapragada (S'12-M'14) received his Ph.D. degree in Mechanical Engineering from the University of Maryland, College Park in 2013. He was a Postdoctoral Scholar in the Department of Mechanical and Aerospace Engineering at the University of California, San Diego from 2014 to 2017. He is currently an Associate Professor in the Department of Electrical Engineering at the Indian Institute of Science, Bangalore, India. His research interests include networked control systems, distributed control,

multi-agent systems and dynamics of social systems.